Author's response to reviews

Title: Incidence and characteristics of invasive Staphylococcus aureus infections: a population-based study in 3 regions of Korea, 2009-2011.

Authors:

Kyoung-Ho Song (khsongmd@gmail.com)
Eu Suk Kim (yonathan@hanafos.com)
Hye-yun Sin (angel-sin@hanmail.net)
Kyung-Hwa Park (medkid@dreamwiz.com)
Sook-In Jung (sijung@cnu.ac.kr)
Nara Yoon (shinenara@gmail.com)
Dong-Min Kim (drongkim@chosun.ac.kr)
Chang Seop Lee (lcsmd@chonbuk.ac.kr)
Hee-Chang Jang (haroc153@naver.com)
Yoonseon Park (yoonseony@gmail.com)
Kkot Sil Lee (mdflorid@empal.com)
Yee Gyung Kwak (philmed202@hanmail.net)
Jae Hoon Lee (john7026@wku.ac.kr)
Seong Yeon Park (psy99ch@hanmail.net)
Minkyo Song (mksong@snu.ac.kr)
Sue K Park (suepark@snu.ac.kr)
Yeong-Seon Lee (yslee07@nih.go.kr)
Hong Bin Kim (hbkimmd@snu.ac.kr)

Version: 2 Date: 20 September 2013

Author's response to reviews:

As you indicated, we have added the following ethics statement in the Methods section of the manuscript.

"This study was approved by the institutional review board of Seoul National University Hospital and Seoul National University Bundang Hospital in compliance with the Helsinki Declaration."

Thank you for your consideration.